

MBE, 20 (4): 7113–7139. DOI: 10.3934/mbe.2023307 Received: 19 September 2022 Revised: 09 November 2022 Accepted: 15 December 2022 Published: 10 February 2023

http://www.aimspress.com/journal/MBE

### Research article

# Effects of inter-industry agglomeration on environmental pollution:

# **Evidence from China**

# Li Xu<sup>1</sup>, Ping Guo<sup>1</sup> and Guoqin Pan<sup>2,\*</sup>

- <sup>1</sup> School of Economics and Trade, Hunan University, Changsha 410006, China
- <sup>2</sup> School of Economics, Nankai University, Tianjin 300071, China
- \* Correspondence: Email: guoqinpan@126.com.

**Abstract:** Industrial pollution comes not only from within industries, but also from between industries that are strongly linked. From the perspective of agglomeration, this study explores the mutual transmission of pollution between different manufacturing industries. We found that there is an inverted U-shape relationship between inter-industry agglomeration and environmental pollution among 20 Chinese manufacturing industries. Energy intensity, which is an important transmission path from agglomeration to pollution, is positively related to the energy consumption of industries with some degree of agglomeration. Besides, the expansion of production scale caused by inter-industry agglomeration leads to more energy consumption and pollution. Furthermore, the innovative technology resulting from inter-industry agglomeration reduces environmental pollution but does not have a significant impact on energy consumption.

Keywords: inter-industry agglomeration; environmental pollution; scale effect; energy; innovation

### 1. Introduction

With the acceleration of industrialization worldwide, collaboration among industries is deepening and the effects of inter-industry agglomeration between different industries are becoming increasingly complex. On the one hand, the development of industrialization promotes the penetration of technological resources and optimizes the allocation of resources. On the other hand, the division of labor and cooperation among various industries, as well as the expansion of the productive scale, increase the severity of inter-industrial pollution. Therefore, understanding the relationship between inter-industry agglomeration and environmental pollution problems is extremely important. It is undeniable that the rapid growth of China's economy in the past has been partly driven by high energy consumption, which has resulted in high pollution [1]. According to British Petroleum's Energy Statistics Yearbook 2020, China accounts for 30.7% of global carbon emissions, making it one of the largest carbon emitters. SO<sub>2</sub>, nitrogen oxides, dust, dust, industrial wastewater and other pollutant emissions also rank among the top in the world. In recent years, the frequent occurrence of environmental events has been pushing governments to pay much more attention to the environmental pollution problem. In 2020, the total investment in environmental pollution control nationwide is 1063.89 billion RMB, of which the total direct investment in environmental pollution control.

The current research mainly focuses on the agglomeration between manufacturing and services to reduce transaction costs [2,3]. For example, service industries tend to locate near manufacturing industries, thus forming an inter-industry agglomeration. Industrial agglomeration can improve the environment, strengthen environmental regulation and alleviate the "pollution paradise" problem [4]. Government-dominated industries only significantly promote local ecological environmental pollution control, while market-driven inter-industry agglomeration can also promote ecological environmental pollution control in the surrounding region through its spatial spillover effect [3]. Due to the exchange of technology and knowledge, the manufacturing sector gathers to expand the production scale, resulting in environmental pollution problems. Therefore, as compared to previous studies, we pay more attention to the cluster between different manufacturing industries, or inter-industry agglomeration.

In this study, we aimed to elucidate the relationship between inter-industry agglomeration and environmental pollution. Also, we endeavored to understand the mechanism behind pollution caused by agglomeration across industries, laying a foundation for the realization of energy conservation and emission reduction in the industrial industry. Based on the measure of inter-industry agglomeration for 20 Chinese manufacturing industries<sup>1</sup>, we found that there is an inverted U-shaped relationship between inter-industry agglomeration and pollution. Energy intensity is an important transmission path from agglomeration to pollution, and it is positively related to industrial energy consumption and inter-industry agglomeration. The expansion of the production scale brought about by the inter-industry agglomeration caused by inter-industry agglomeration does not reduce energy consumption, but has a significant negative impact on environmental pollution.

The contributions of this paper are as follows. First, the pollution is not caused by a single manufacturing industry. Under the backdrop of the professional division of labor, the agglomeration among industries has deepened, and environmental pollution is not only caused by its own production, as it is also related to other industries with close cooperation. The energy consumption and pollution would be different due to the different agglomeration schemes among industries, which reflects the economic connection between production and operation among different industries based on agglomeration. Second, we attempted to put the manufacturing inter-industry agglomeration, energy intensity and environmental pollution in a unified research framework. This paper analyzes the effects of manufacturing inter-industry agglomeration can play a role in energy conservation and emission reduction, and whether energy intensity can serve as a bridge between saving energy and reducing pollution. Furthermore, we analyze the changes in environmental pollution caused by inter-industry

<sup>&</sup>lt;sup>1</sup> According to "The Industry Classification Standards of National Economy Industry Classification and Code" (2002)

agglomeration from the perspective of energy intensity, which can reveal the mechanisms behind transmission paths from agglomeration to pollution across different industries.

The rest of paper study proceeds as follows: Section 2 provides a literature review; Section 3 introduces the theoretical analysis; Section 4 shows the empirical method and data; Section 5 presents the empirical results and potential mechanisms; Section 6 concludes the paper.

### 2. Literature

#### 2.1. Industrial agglomeration and environmental pollution

The concept of inter-industry agglomeration was first proposed by Ellison and Glaeser [5], and it is mainly manifested in the correlation between different industries and the existence of overlap in their spatial layouts [6]. In essence, coagglomeration is a phenomenon resulting from inter-industry agglomeration. At present, scholars mainly focus on coagglomeration between the manufacturing industry and the producer services industry [2,3,7], frequently adopting the perspective of industrial spatial layout to measure the coagglomeration [8]. To reduce transaction costs, service industries tend to locate near manufacturing industries, thus forming an inter-industry agglomeration. Government-dominated industries only significantly promote local ecological environmental pollution control, while market-driven inter-industry agglomeration can also promote ecological environmental pollution control in the surrounding region through its spatial spillover effect [3,9]. Ellison and Glaeser [5] observed the nature of the agglomerative force, exploring it through the agglomeration of related industries and those with strong upstream and downstream linkages.

There are few studies on the impact of inter-industry agglomeration on environmental pollution, and most scholars are committed to examining the impact of industrial agglomeration on environmental pollution. On the one hand, industrial agglomeration can offer positive externalities that reduce environmental pollution. Because agglomeration is based on the specialized division of labor, specialized agglomeration promotes inter-industry technology exchange and consumes less energy; thus, it is characterized by positive externalities [10]. Industrial agglomeration reduces environmental pollution mainly by reducing resource mismatch [11], forming scale effects [12], promoting industrial structure adjustment, producing spillover effects and driving innovation [13-18]. Industrial agglomeration can improve the environment, further strengthen environmental regulation and alleviate the "pollution paradise" problem [4]. Sun et al. [19] believe that the central government should reduce market fragmentation, promote market integration, improve resource allocation efficiency and improve environmental quality [20]. Industrial agglomeration can promote economic growth, technological progress and environmental awareness [21–24], thus improving regional environmental pollution. Glaser and Kahn [25] believes that economic agglomeration effectively reduces the commuting distance and traffic pollution emissions, implying that cities are more economical and environmentally friendly than villages. Zeng and Zhao [4] included cross-sector and trans-boundary pollution in their new economic geography model, and their research shows that manufacturing agglomeration can reduce the "pollution paradise" effect to the point of non-existence given strict environmental regulations. Environment-oriented technological progress can promote the improvement of environmental quality and realize coordinated development between the economy and the environment.

On the other hand, agglomeration also represents a negative externality that affects the environment. The expansion of production caused by industrial agglomeration leads to more serious environmental pollution problem through the increase of energy consumption. Brakman [26] proposes that agglomeration has a congestion effect, and they found that it could have a negative impact, such

as environmental pollution, limited resources and transportation problems. Frank [27] pointed out that agglomeration is the main factor leading to high levels of air and water pollution. Industrial agglomeration leads to shortages in land supply, aggravates the development of land resources and causes pollution that reduces water quality [28]. The concentration of environmental pollution is related to the energy rebound strength, and the energy rebound strength is closely related to the economic development stage [29]. An extensive economic development mode causes a stronger energy rebound effect to occur, and technological innovation in this mode does not reduce pollution emissions, but would produce technology in the "energy-saving, low-efficiency trap", which is not conducive to optimizing the environment. Wang et al. [30] believe that the entry of new enterprises is an important factor leading to the scale expansion of the development zone, and that environmental pollution is also due to the introduction of polluting enterprises. There is a two-way causal relationship between environmental pollution and agglomeration, i.e., agglomeration intensifies pollution emissions, while pollution limits industrial agglomeration to a certain extent. The energy rebound effect caused by technological changes reduces the cleaning effect brought about by the energy structure and aggravates haze pollution. Liu et al. [31] have shown that industrial agglomeration intensifies the level of environmental pollution, although the negative environmental effect of industrial agglomeration began showing a weakening trend after China's economy entered the new normal.

However, some scholars believe that the impact of agglomeration on the environment has a dual effect and is not simply either a one-way promotional or inhibitory effect. Erik and Peter [32] found that there is a nonlinear relationship between industrial agglomeration and pollution in China's industry. Fan and Scott [33] discuss agglomeration in the less-developed countries of East Asia and its relationship with economic development and growth. Lan, Kakinaka and Huang [34] support the research conclusion of nonlinearity at the urban level. Zhu and Xia [35] used a general equilibrium model, selected a new city as the threshold variable and verified the inverted U-shaped curve relationship between industrial agglomeration and pollution. Liu et al. [31] divided the data into two stages before and after the "new normal"; they found that the negative externalities of the agglomeration environment gradually weaken. Ye and Yu [36] studied industrial agglomeration and haze pollution across China, including the central and eastern regions, finding that they maintain a typical inverted U-shaped curve relationship.

# 2.2. Industrial agglomeration and pollution

Energy consumption is the most direct factor affecting environmental pollution. At present, considering the relationship between agglomeration and energy intensity, scholars prefer to improve energy efficiency through technological progress [37,38]. Increasing levels of localized knowledge spillovers and the substitution of internal inputs with external inputs may lead to fewer errors in decision-making, as well as in the execution of production tasks, causing firms to become technically more efficient relative to the production frontier. Naik and Bagodi [39] believe that the prevailing technology and a lack of compatibility with new technology are barriers to energy efficiency improvement. Yi et al. [40] pointed out that energy-saving technology advances cannot effectively reduce smog pollution due to the energy rebound effect. Tveteras and Battese [41] found econometric support for positive agglomeration externalities for both the production frontier and technical inefficiency. Ciccone and Hall [42] studied changes in the role of agglomeration externalities in the industry life cycle. According to their research, because the competitive mode, innovation and learning opportunities of the industry change over time [43–45]. Martin et al. [46] discussed whether the

agglomeration economy originates from specialization (within industries), diversity (between industries) or overall density. Research supports the idea that industrial clusters located within diverse and dense cities can enhance productivity. Only industrial agglomeration can have a positive impact on industrial energy efficiency [47]. The regional development policy implemented by the state has slowed down the pace of energy efficiency improvements in China's paper industry. The government needs to take regional characteristics into consideration and consciously guide industrial enterprises to concentrate in advantageous areas. Tahir and Ahmad [48] argue that energy intensity may decline with economic growth due to the technological changes that accompany growth. Technological innovation [49,50] is not only able to reduce regional haze pollution, but it can also indirectly reduce haze pollution in neighboring provinces through the knowledge spillover effect [51]. Provincial energy efficiency has a significant spatial spillover effect [52,53], and agglomeration has a U-shaped effect on energy efficiency. Rahman and Alam [54] explored the relationship between health status and health expenditure (public and private), energy consumption and environmental pollution in the SAARC-BIMSTEC region.

In the energy and environmental pollution research, most scholars focus on analyzing the energy consumption structure and believe that it is closely related to environmental quality. Both resource extraction and environmental pollution have significant inhibitory effects on regional long-term economic development. Environmental pollution is the main source of the resource curse. Sun et al. [55] established a new indicator system for the relationship between China's economic and social development, natural resource consumption and environmental pollution. The combustion of coalbased fossil energy directly causes air and water pollution and aggravates environmental deterioration. Some of the results show that servitization leads to an improvement in energy consumption, which thus improves environmental performance [56]; alternatively, Faisal and Afra [57] examined the relationship between energy consumption and carbon emissions from the perspective of economic growth. With economic development, carbon emissions show a growth trend [58,59] with the relationship chain, and economic growth also increases. Li and Xu [60] pointed out that the impact of investment bias [61–63] and a factor market [64–66] should be considered when formulating energy conservation and emission reduction policies. Different industries have different emission reduction potentials [67] and energy-saving potentials. Baydoun and Aga [68] show that economic growth and energy consumption decrease environmental sustainability, while globalization improves it. There is a one-way causality from energy consumption and globalization to CO<sub>2</sub> emissions in Gulf Cooperation Council (GCC) countries. According to the findings, environmental pollution in GCC countries is output-driven, which means that it is determined by the amount of energy generated and consumed. This is consistent with the conclusion of Liu et al. [69]. Energy consumption is the main factor leading to ecological pollution. Some scholars estimated the energy rebound effect in China's industrial sector [70,71]. Output growth is the main factor driving industrial energy consumption, while energy intensity reduction and structural change play leading roles in moderating industrial energy consumption. Ushifusa and Tomohara [72] pointed out that energy intensity has an inverted U-shaped impact on carbon emissions, and that it plays an intermediary role in the economic agglomeration and pollution emissions. Li [22] analyzed the impact of the energy rebound effect on carbon emissions in different urbanization stages. The impact of the short-term energy rebound effect on carbon emissions is greater than that of the long-term energy rebound effect [73], and both decrease with an increase in the urbanization rate.

Adjusting the energy consumption structure appears to be an effective way to alleviate serious pollution. Starting from the internal structure of energy consumption, Mudakkar et al. [74] divided energy into electrical energy, fossil energy and nuclear energy and explored the differences in the

environmental impact of the energy consumption structures in different industries. The long-term governance of air quality depends upon the improvements to the energy consumption structure. In the short term, it is necessary to reduce the use of coal to alleviate severe ecological problems. To adjust the energy structure, the industrial structure also needs to be adjusted. The long-term strategy for achieving a green environment is to adjust the energy structure and the internal structure of the industry. Liu and Xiao [75] further verified through empirical research that the energy structure is the main factor affecting carbon emissions, emphasizing that regions with coal consumption as the main energy source should explore the use of emission reduction methods in their energy consumption structure to reduce environmental pollution. There is no environmental Kuznets curve (EKC) for CO<sub>2</sub> emissions from economic growth, nonrenewable energy production or foreign trade in China. However, Chen et al. [76] found that the inverted U-shaped hypothesis of the EKC is supported in the long run after the renewable energy production variable is added. Administrative power leading the allocation of resources creates obstacles in the flow of factors between regions, aggravates the market segmentation of resource-based cities and is not conducive to knowledge spillover and pollution control technology sharing [77,78].

Existing literature on the relationship between industrial agglomeration, energy intensity and environmental pollution is relatively rich, but the following points can be improved. First, all studies focus on the agglomeration of a single industry but do not pay attention to the coagglomeration among different types of industries. Through the division of labor and cooperation or the coagglomeration of upstream and downstream industries in the industrial chain, the impact of inter-industrial agglomeration on energy demand and pollution emission is also affected by the industrial type. Second, there are few studies on inter-industry agglomeration, industrial innovation and environmental pollution under the same theoretical framework. Inter-industry agglomeration brings economies of scale through centralized regulation and technology spillover effects, affects energy consumption and ultimately affects industrial pollution emissions. Internal relationships exist among the three, and putting them in the same framework allows for a more comprehensive understanding of the impact of inter-industry agglomeration on pollution. In this study, based on the existing literature, we improve the related research on environmental pollution by addressing the above two shortcomings, thereby providing valuable results that can be used to support environmental governance.

# 3. Theoretical analysis

The impact of inter-industry agglomeration on environmental pollution is mainly related to the transmission of energy intensity, and there is a positive correlation between energy intensity and environmental pollution, so we should pay more attention to the impact of inter-industry agglomeration on energy intensity. With economic growth, energy consumption and industrial agglomeration have increased significantly. Industrial agglomeration and energy have a significant "inverted U-shaped" relationship [79]. In the early stage of industrial development, the scale effect [80] caused by inter-industry agglomeration would lead to the increase in energy consumption, resulting in more environmental pollution. With the development of industry, technology, labor and capital continue to interact between industries [81,82]. There is a spillover effect [53,83,84] among industries through the division of labor and cooperation between industries, as all types of public infrastructure are built and some enterprise clusters form. Inter-industry agglomeration could reduce the cost of production and transportation and increase the exchange of technology and knowledge. The manufacturing industries are gradually shifting toward low-energy consumption and low-pollution industries, reducing energy

consumption and pollutant emissions [85]. Therefore, the influence of inter-industry agglomeration on energy intensity presents as an inverted U-type relationship, and it further affects environmental pollution. Hypothesis 1 is proposed.

Hypothesis 1: The impact of inter-industry agglomeration on the environment is represented by an inverted U-type relationship through the path of energy intensity.

Inter-industry agglomeration can lead to technology improvement because technology connectivity shares public technology or overflows among industries [84]. Therefore, inter-industry agglomeration would promote the reduction of pollution emissions through technological upgrades. Manufacturing coagglomeration brings about technological upgrades and uses advanced equipment to reduce pollution emissions [86]. Because the government collects emission fees on enterprises [87,88], enterprises pay more attention to emission reduction effects rather than energy-saving effects. Furthermore, technological progress caused by inter-industry agglomeration would lead to the expansion of the production scale [89], which would increase energy demand. The new technologies adopted by enterprises are often used to reduce emissions, but they cannot effectively reduce energy demand by the scale of production [90]. The coagglomeration of manufacturing industries brings technological progress [91,92], prompting enterprises to be more motivated to expand the production scale, which leads to an increase of energy demand. Due to the continuous expansion of the industrial scale, the new technology would have difficulty offsetting the increased energy demand caused by scale expansion, resulting in more energy consumption with the new technology than that without the new technology [93]. Therefore, we propose Hypotheses 2a and 2b.

Hypothesis 2a: The technology innovation induced by inter-industry agglomeration can reduce environmental pollution.

Hypothesis **2b: The expansion of the production scale brought about by coagglomeration** has a negative effect on environmental pollution.

#### 4. Research design

#### 4.1. Model settings

Inter-industry agglomeration impacts the environmental pollution caused by other related industries in the process of production, transportation and exchange. We focus on the coagglomeration of 20 manufacturing industries (Table 1). This paper discusses the impact of coagglomeration on pollution emissions. The model is as follows:

$$M_{it} = \beta_0 + \beta_1 W_{ijt} M_{jt} + \beta_2 (W_{ijt} M_{jt})^2 + \beta_3 z_{it} + u_i + v_t + \varepsilon_{it}$$
(1)

$$y_{it} = \lambda_0 + \lambda_1 M_{it} + \lambda_2 z_{it} + u_i + v_t + \varepsilon_{it}$$
<sup>(2)</sup>

where  $W_{ij}$  indicates the coagglomeration level of industry *i* and industry  $j^2$ . The coagglomeration that needs to be noted is represented by an  $n \times n$  matrix, and the other indicators are given in an  $n \times 1$  matrix.

<sup>&</sup>lt;sup>2</sup>  $W_i$  is the agglomeration of the same type of manufacturing industry *i*, and  $W_{ij}$  indicates the agglomeration of different types of manufacturing industries (industry *i* and industry *j*). Therefore,  $W_i \times M_{it}$  and  $W_{ij} \times M_{jt}$  have different meanings.  $W_i \times M_{it}$  indicates the synergy of an industry's agglomeration and the energy intensity of the industry.  $W_{ij} \times M_{jt}$  indicates the synergistic effect of the agglomeration between two manufacturing industries (industry *i* and industry *j*) and the energy intensity of the manufacturing industry *j*.

Due to the particularity of the coagglomeration index, we cannot observe the impact of coagglomeration on environmental pollution separately. Considering that energy intensity is a mechanism between coagglomeration and the environment [94], we studied the impact of the interaction between coagglomeration and energy intensity on the environment. That is, the energy intensity of industry *i* is affected by the energy intensity of others converging with it; it is used to obtain the impact of coagglomeration on environmental pollution by analyzing the energy intensity.  $y_{it}$  is the emissions of pollutants for industry *i* during period *t*. We chose to focus on wastewater emissions, sulfur dioxide emissions, carbon dioxide emissions, smoke (dust) emissions, solid waste generation and comprehensive indicators of environment<sup>-1</sup> pollution when estimating the impact of coagglomeration, openness, ownership and Research  $e^{-A}$  development strength (R&D) inten<sup>-i+</sup>y.  $M_{ij}$  is the energy intensity of industry *j* during period *t*,  $u_i$  represents the industry fixed effect,  $v_t$  represents the time fixed effect and  $\varepsilon_{it}$  indicates the random interference term.

		8
Industry code and name	Industry code and name	Industry code and name
C13 Agricultural and sideline food processing industry	C26 Chemical raw materials and chemical products manufacturing	C35 General equipment manufacturing
C14 Food manufacturing	C27 Pharmaceutical manufacturing	C36 Special equipment manufacturing industry
C15 Wine, beverage and refined tea manufacturing	C28 Chemical fiber manufacturing industry	C37 Transportation equipment manufacturing industry
C16 Tobacco products industry	C31 Furniture manufacturing and nonmetal mineral products	C39 Electrical machinery and equipment manufacturing industry
C17 Textile industry	C32 Ferrous metal smelting and calendering industry	C40 Computer, communication and other electronic equipment manufacturing industry
C22 Paper and paper products industry	C33 Nonferrous metal smelting and calendering industry	C41 Instrumentation and cultural manufacturing
C25 Petroleum processing, coking and nuclear fuel processing industries	C34 Metal products industry	

Table 1. Classification of manufacturing industry.

Notes: Two-digit and 20 sub-industries were selected based on Industry Classification and Code of the National Economy. The reason for choosing these sectors is that, first, they cover various factor-intensive industries, representing the overall level of manufacturing development. Second, these industries are not heavily dependent on natural resources, which means that they can be transferred between regions and will have a strong scale effect.

Formula (1) shows that the energy intensity of industry *i* is affected by the energy intensity of industry *j* in the case of inter-industry agglomeration.  $\beta_1$  indicates that the energy intensity of industry *i* is affected by the inter-industry agglomeration of industry *i* and industry *j* and the energy intensity of industry *j*.  $W_{ij}M_{ji}$  indicates the effect of the synergy between the two industries (industry *i* and industry *j*) and the energy intensity of another agglomeration industry *j*. Formula (2) reveals that the energy efficiency of the industries affects the level of energy consumption and, then, of pollution emissions. Combining Formulas (1) and (2), it can be observed that the pollution emissions of industry *i* are

affected by the energy intensity of industry j at a certain level of inter-industry agglomeration, which indicates that the environmental pollution of industry i is also affected by the energy consumption of related agglomeration industries.

#### 4.2. Index selection and data source

#### 4.2.1. Index selection

Explained variable. The main methods for measuring environmental pollution are the factor synthesis method, principal component analysis method, entropy method and comprehensive index method, among others. In this study, the entropy method [95] was used to calculate the environmental pollution index, which includes SO<sub>2</sub> emissions, smoke or dust emissions, CO<sub>2</sub> emissions<sup>3</sup>, wastewater emissions and solid waste production. Then, the principal component analysis method [97] was used to measure the environmental pollution index for the robustness test. CO<sub>2</sub> emissions were calculated by using Formula (3).

Outcome variable. We were interested in capturing the degree of energy intensity [94] and that of manufacturing coagglomeration. The detailed data sources and the process for computing the interindustry agglomeration index are as follows:

(1) The total industrial output value for all regions and industries. The index takes 1990 as the base period for the gross industrial output value for all regions and industries and adjusts it according to the factory price index of industrial producers. The data were taken from the China Industrial Economic Statistical Yearbook from 2000 to 2016.

(2) The number of enterprises in different types of industries. Due to the lack of data on the number of enterprises in various industries in 2005, we used the averaging method to fill the data gaps. Because the micro data of the national industrial database are only available through 2013, the H index was only calculated to 2013. To ensure the integrity of the data, the H index values from 2014 to 2016 were predicted by exponential smoothing.

(3) The total output value for each region. The gross output value of each region was measured by GDP (100 million rmb), and the data were taken from the China Statistical Yearbook (2000–2016).

Control variables. Environmental regulation is also an important factor affecting environmental pollution. We referred to Lin et al. [96] to measure the comprehensive utilization rate of waste. When the environmental regulation intensity increases, enterprises are afraid to discharge too many pollutants and the comprehensive utilization rate of waste is high; otherwise, the utilization rate is low. R&D intensity is measured by internal and external R&D expenditures [98]. According to endogenous economic growth theory, the higher the R&D investment, the better the technological innovation and progress, which causes corresponding improvements in the industrial production efficiency [99] and green economic efficiency. Therefore, the greater the R&D intensity, the more likely that environmental pollution emissions would be reduced. The degree of opening up is determined by the shipping ratio [100]. Opening up is conducive to attracting foreign investment and green technology to support energy conservation and emission reduction. In addition, the following control variables were selected at the industrial level: ownership structure is represented by the proportion of stateowned enterprises [96], as ownership is found to be an important factor affecting the industrial

<sup>&</sup>lt;sup>3</sup> Carbon dioxide emissions from the manufacturing production process are mainly related to the combustion of chemical fuels. For example, manufacturing industries, such as the metal smelting, oil processing, coking and nuclear fuel processing industries, produce a large amount of carbon dioxide in the production process, leading to abnormal climate, elevated sea levels, melting glaciers and reduced numbers of animals and plants, which has a negative impact on the natural environment. Therefore, we include carbon dioxide in the category of pollutants.

structure; the population of the employed labor force was selected for the scale, noting that the output levels of each industry are different because of productive scale; and the energy consumption structure is represented by the proportion of coal consumption in total energy consumption [101]. The cost structure determines the volume of pollution emissions.

#### 4.2.2. Measures of relevant indicators

#### A. Carbon emissions

In the industrial production process, environmental pollution is caused by energy consumption. To objectively reflect the pollution situation of various industries, we selected the following five types of pollutants: sulfur dioxide emissions, smoke or dust emissions, carbon dioxide emissions, wastewater emissions and solid waste generation. However, although carbon dioxide emissions are the main indicators of greenhouse gases, the relevant data are not provided in the relevant yearbook. According to the research of Qu et al. [101], carbon dioxide emissions can be calculated by multiplying the energy consumption by the carbon dioxide emission coefficient of the focal industry; the formula is as follows:

$$\sum_{i=1}^{n} CO_{2i} = \sum_{i=1}^{n} E_i \times NCV_i \times CEF_i \times COF_i \times (\frac{44}{12})$$
(3)

where *n* refers to the types of energy consumption in the industry. In this study, the following eight sources of consumed energy were selected: coke, coal, kerosene. gasoline, fuel oil, natural gas, diesel oil and crude oil.  $E_i$  is the *i*-th source of energy consumed,  $NCV_i$  is the average calorific value of the *i*-th energy source,  $CEF_i$  is the carbon content per unit calorific value of the *i*-th energy source and  $COF_i$  is the oxidation level of the *i*-th energy source. The molecular weights of CO<sub>2</sub> and carbon are 44 and 12, respectively. The CO<sub>2</sub> emissions of the *n*-th energy source can be calculated by using Formula 4, and the total CO<sub>2</sub> emissions of various industries can be obtained by summing the *CO*<sub>2</sub> emissions of all *n* energy sources.

#### B. Agglomeration between different manufacture industries

Based on the calculation method of Ellison and Glaeser [5], we considered the scale of enterprises in various industries, the market structure of industries and other factors as factors influencing interindustry agglomeration. The calculation formula is as follows:

$$W_{ij} \equiv \frac{[G_i/(1-\sum_i^M x_r^2)] - \sum_i^M x_i^2 H_i - \sum_i^{i=20} W_i x_i^2 (1-H_i)}{1 - \sum_i^{i=20} x_i^2}$$
(4)

where  $W_{ij}$  represents the coagglomeration of different types of manufacturing industries and  $W_i$  represents the agglomeration within the industry and the same industry type. Therefore, Formula (5) represents intra-industry agglomeration, which is different from Formula (4).  $H_i$  said that the *i*-th industry *H* index is the enterprise summary of industry data.  $G_i$  is the spatial Gini coefficient. The specific calculation is as follows:

$$W_{i} \equiv \frac{G_{i} - (1 - \sum_{i}^{M} x_{r}^{2}) H_{i}}{(1 - \sum_{i}^{M} x_{r}^{2})(1 - H_{i})}$$
(5)

$$G_i = \sum_{i=1}^{M} (q_{ir} - x_r)^2$$
(6)

$$H_i = \sum_{j=1}^{N_i total} Z_p^2 \tag{7}$$

 $q_{ir}$  represents the GDP of the *i*-th industry in the *r*-th region in that year, and it is deflated by the industrial producer price index (1990 = 100);  $x_r$  represents the gross domestic product of the *r*-th region as a proportion of the national gross domestic product;  $Z_p$  represents the employment number of the *p*-th enterprise as a proportion of the total employment, which measures the relative scale of the enterprise in the industry. The data are at the enterprise level (a total of 161671 enterprises). *M* represents the number of provinces and cities in China.

#### 4.2.3. Data sources

CO<sub>2</sub> emission can be calculated by using Formula (3). The average heat generation (*NCV<sub>i</sub>*) data for various energy sources were taken from the General Rules for Comprehensive Energy Consumption Calculation (GB/T 2589-2008). The carbon content per unit calorific value ( $^{CEF_i}$ ) and oxidation rate ( $^{COF_i}$ ) of various energy sources were taken from the Intergovernmental Panel on Climate Change (IPCC) national greenhouse gas inventory guidelines. The energy consumption data were taken from the China Energy Statistical Yearbook. The relevant indicators of industrial wastewater, waste gas and waste were taken from the China Environmental Database; other data were taken from the China Industrial Economic Statistics Yearbook (1999–2016).

When we calculate inter-industry agglomeration, the spatial Gini coefficient needs to be measured first (Formula (6)). The data were used to calculate the Gini coefficients from the China Industrial Economic Statistical Yearbook. We calculated the Heffendar index by using Formula (7); the data were derived from the statistical yearbook of China's industrial enterprises and the statistical yearbook of China's industrial economy. Finally, we calculated the level of intra-industry agglomeration (Formula (5)) and the level of inter-industry agglomeration (Formula (4)). Controls of the variable data source were taken from the Statistical Yearbook of China Industrial Economy (1999–2016).

### 4.2.4. Results of agglomeration of Chinese industries

We measured the level of coagglomeration among Chinese manufacturing industries from 2000 to 2016 based on the method proposed by Ellison and Glaeser [5]. First, we calculated the intraindustry agglomeration by using Formula (5) and divided into three stages according to Ellison and Glaeser [5]. Three industries (C28, C40 and C41) were calculated to be in the high agglomeration stage, namely, 0.1287, 0.0794 and 0.0912, respectively. The industries in the medium and low intra-industry agglomeration stages account for a large proportion, and the number of manufacturing industries in the medium and low intra-industry agglomeration stages accounts for half of each.

Second, we further calculated the inter-industry agglomeration levels among various industries according to Formula (4). It shows that the inter-industry agglomeration result is a symmetric matrix (Appendix 1, Table A3). From the perspective of the average over the years, C28 and C41 have the highest levels of inter-industry agglomeration with others, all greater than 0.1. First, it shows that the C28 chemical fiber manufacturing industry had the highest synergistic agglomeration level in the whole bisection-digit manufacturing industry, which was as high as 0.2696. In the first half of 2019, the main business revenue of the chemical fiber industry was 428.7 billion yuan, up 7.1% year on year;

the total profit was 13.3 billion yuan. Chemical fibers are made from natural polymer compounds or synthetic polymer compounds. Because of their different chemical compositions, they have different properties and wide uses. For example, adhesive fiber is used as a curtain cloth in industry applications, so it is closely related to the transportation manufacturing industry. Polyester and silicone rubber are used to produce artificial skulls, which are applications that are closer to the medical manufacturing industry; Cotton is used to make fishing nets, toothbrushes and brush-related products for the food manufacturing and textile industries. Acrylic has aging resistance and water absorption properties, which are appropriate for applications in the construction, water conservancy, medical and clothing industries. Therefore, chemical fiber and other industries have the highest levels of inter-industry agglomeration. Second, the C41 instrument, cultural office supplies machinery, other measurement instrument manufacturing industry and other industries reached the second-highest level. These include many industries and are widely used in the industrial manufacturing industry. Instruments can also have automatic control, alarm, signal transmission and data processing functions. For instance, pneumatic adjustment instruments used for the automatic control of industrial production processes, electrical adjustment instruments and distributed instrument control systems are essential in the entire industrial manufacturing process. In addition, the C40 communication equipment computer and other electronic equipment manufacturing industry ranked third in the degree of coordination, and the agglomeration level with other industries ranged between 0.08–0.22. Communication equipment and other electronic equipment serve as the hub of signal transmission in the production of other industries, cooperating with each link and each department, and the industry has a high degree of coordination with others.

# 5. Empirical results

### 5.1. Main results

Inter-industry agglomeration can affect energy intensity through economies of scale effects and technology spillover effects. It drives changes in energy efficiency and ultimately impacts environmental pollution. Therefore, this section is divided into the following two types of results: the first focuses on the impact of inter-industry agglomeration on energy intensity, and the second focuses on the effects of energy intensity on environmental pollution. To more clearly illustrate the impact of inter-industry agglomeration and energy intensity on pollution emissions, the annual mean of the above three indicators has been used.



Figure 1. Average of inter-industry agglomeration and pollution emissions.

Figure 1 shows the interaction between the collaborative concentration (i and j industries) and energy intensity of other industries (j industries); the joint impact on environmental pollution in industry i shows an inverted U-shaped relationship. This indicates that pollution emissions in a certain industry are related not only to the inter-industry agglomeration, but also to the energy intensity of coagglomeration in other industries. Because energy consumption is a direct source of environmental pollution, the greater the energy intensity, the greater the pollution emissions, and the relationship between energy intensity and pollution emissions is positive. Therefore, we used the quadratic term in the model settings, arguing that the joint effect of inter-industry agglomeration and energy intensity showed a U-type relationship with environmental pollution.

Inter-industry agglomeration can lead to environmental pollution caused by production, transportation and technology exchange within and across different levels of industries. Based on the panel data of 20 manufacturing industries from 2000 to 2016, the 2SLS regression model was used. It is worth noting that, when considering the impact of manufacturing inter-industry agglomeration on energy intensity, as well as the impact of energy intensity on pollution, the corresponding explanatory variables lagging 2–3 periods were used as instrumental variables.

We take the result of 2SLS as an example in Table 2. The primary and secondary coefficients for the interaction capturing the agglomeration and the energy intensity of industry j were 1% and 5%, respectively, and an inverted U-shaped curve was formed. The multiplication of the coagglomeration index and the energy strength of other industries was less than 0.649. Energy intensity increases with the energy intensity in others and fails to yield the dividend effect of inter-industry agglomeration under the constraint of the coagglomeration, which is not conducive to the reduction of energy consumption [79]. In the early stage, when the inter-industry agglomeration is low, the industrial chain is not mature, the common technology or information technology is relatively low and, thus, clusters do not have a technology spillover effect. With the deepening of the inter-industry agglomeration, the professional division of labor between manufacturers deepens and the industrial chain is increasingly improving. The scale effect of the coagglomeration is prominent. Besides, coagglomeration enables manufactures with similar or common technologies to achieve technology exchange and improve energy utilization efficiency, and the energy intensity shows a downward trend.

The impact of energy intensity on environmental pollution is shown in Table 3. Energy consumption is a direct factor in environmental pollution. The consumption of fossil fuels represented by coal directly generates  $CO_2$ ,  $SO_2$  and other pollutants, which aggravates environmental pollution. The empirical regression yielded that energy intensity has a significant positive correlation with environmental pollution. Furthermore, considering how energy intensity does indeed aggravate environmental pollutants, the energy intensity of industry *i*, at a significance level of 10%, has a negative correlation with industrial wastewater, industrial SO<sub>2</sub> and soot emissions of other industries. Generally, industrial wastewater and smoke and dust emissions are mostly non-energy-dependent industries, with low energy demand. When the production of energy-dependent products is increased, it naturally occupies the production space of non-energy-dependent products and reduces the emission of pollutants such as wastewater and soot. However, energy intensity has a significant positive correlation with the total amount of industrial solid waste and carbon emissions, indicating that an increase in energy intensity can bring an increase in solid waste and carbon emissions for other industries, thus aggravating environmental pollution.

Variables	Energy-2SLS (M <sub>j</sub> )	Energy-FE (M <sub>j</sub> )
$W_{ij}*M_j$	0.959***	1.356**
	(0.247)	(0.589)
$(W_{ij}*M_j)^2$	-0.739**	-2.270**
	(0.301)	(1.014)
Regulation	-0.01	-0.001**
-	(0.015)	(0.000)
Opening	0.064***	0.170*
	(0.022)	(0.084)
R&D	-0.183**	-0.112**
	(0.092)	(0.048)
Scale	-0.045**	0.061
	(0.018)	(0.112)
Constant	0.064*	-0.019
	(0.034)	(0.048)
Year FE	Yes	Yes
Industry FE	Yes	Yes
Observations	300	340
R-squared	0.955	0.275

**Table 2.** Impact of coagglomeration on energy intensity.

Notes: The standard errors are in parentheses; \*\*\*, \*\* and \* represent significance at the 1%, 5% and 10% levels. The regression R-value generally reflects the presence of unobservable factors in the economic models and does not affect the causal identification effect. The magnitude of the R-value is not used as a criterion for the judgment of the model. For example, the value of the R-value in Column 3 of Table 2 is 0.275, which does not affect the explanations of the coefficients in the study.

Table 3. Impact of energy intensity on environmental pollution.

Variables	Comprehensive pollution	Wastewater	$SO_2$	Solid waste	Smoke or dust	CO <sub>2</sub>
$M_i$	1.443***	-0.730***	-0.930**	2.105***	-1.332***	1.318***
	(0.362)	(0.226)	(0.426)	(0.444)	(0.422)	(0.259)
Regulation	-0.0004	-0.0001	0.0007	-0.0005	0.0002	-0.0001
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Opening	-0.276***	0.046	0.250**	-0.254*	0.245**	-0.041
	(0.107)	(0.067)	(0.125)	(0.131)	(0.124)	(0.076)
Ownership	-0.251*	0.118	0.395***	-0.258	0.324**	0.044
	(0.128)	(0.080)	(0.151)	(0.157)	(0.150)	(0.092)
R&D	0.745*	-0.457*	-0.657	0.139	-0.911*	0.128
	(0.416)	(0.260)	(0.489)	(0.510)	(0.484)	(0.297)
Structure	0.290***	-0.010**	-0.227**	0.145	-0.215**	0.047
	(0.076)	(0.047)	(0.089)	(0.093)	(0.088)	(0.054)
Constant	-0.127	0.412***	0.212**	-0.132	0.258**	-0.063
	(0.091)	(0.057)	(0.107)	(0.111)	(0.106)	(0.065)
Observations	320	320	320	320	320	320
R-squared	0.349	0.961	0.866	0.590	0.710	0.914
Year FE	yes	yes	yes	yes	yes	yes
Industry FE	yes	yes	yes	yes	yes	yes

Notes: The standard errors are in parentheses; \*\*\*, \*\* and \* represent significance at the 1%, 5% and 10% levels.

7126

Combining Tables 2 and 3, the primary and secondary coefficients for the interaction between inter-industry agglomeration and energy intensity were calculated to be 1.3838 and -1.0664, respectively. This shows that the pollution emission level of industry *i* is closely related to the energy consumption of other industries, and that, as the degree of inter-industry agglomeration deepens, the pollution emission level first rises and then declines. When the inter-industry agglomeration reaches a certain value, the energy intensity of industry *j* affects the pollution emission of industry *i*. Therefore, to achieve the goal of reducing pollution, an industry must also consider the energy efficiency and energy consumption of other industries that are closely related to its industrial production, division of labor and exchange.

### 5.2. Robustness test

### 5.2.1. Comprehensive index of environmental pollution

We selected five types of pollutants and then used the entropy method to fit the comprehensive index of environmental pollution. To avoid biased research issues caused by different research methods, we further used principal component analysis for verification. The results are shown in Table 4. Columns (1) and (2) show the increased significance of the impact of inter-industry agglomeration on the environment, indicating that energy intensity is the intermediate mechanism between coagglomeration and environmental pollution.

Variables	Energy (M <sub>i</sub> )	Main component	Drop CO <sub>2</sub>
Mi		0.076***	0.418***
		(0.006)	(0.037)
$W_{ij}*M_j$	0.959***		
	(0.247)		
$(W_{ij}*M_j)^2$	-0.739**		
	(0.301)		
Constant	0.064*	0.002	0.001
	(0.034)	(0.003)	(0.002)
Controls	Yes	Yes	Yes
R-squared	0.955	0.943	0.928
Year FE	yes	yes	yes
Industry FE	yes	yes	yes
Observations	300	320	320

### Table 4. Impact of industrial agglomeration on pollution emissions.

Notes: The standard errors are in parentheses; \*\*\*, \*\* and \* represent significance at the 1%, 5% and 10% levels.

#### 5.2.2. Removal of the greenhouse gas CO<sub>2</sub>

There is some dispute about whether  $CO_2$  is an environmental pollutant. In the previous section,  $CO_2$  is included in the comprehensive index of pollutants. To consider this alternative perspective, we reconstructed the comprehensive index of environmental pollution by removing  $CO_2$ . The results are shown in Table 6. Columns (1) and (3) show the results for the comprehensive impact of the coordinated agglomeration of two different industries on environmental pollutants through the

transmission of energy intensity; Column (5) is the impact of the interaction of intra-industry agglomeration and energy intensity on the environment; the results excluding  $CO_2$  emissions as a pollutant are consistent with the calculation results, indicating that the environmental pollution of manufacturing industries is not related only to their own energy intensity. The conclusion is consistent with the previous conclusion.

### 5.3. Mechanisms

### 5.3.1. Impact of the productive scale on environmental pollution

In the relationship between coagglomeration and the environment, the amount of pollution emissions is related not only to the inter-industry agglomeration, but also to the intra-industry agglomeration and the scale [84]. We further analyzed the impact of two factors on environmental pollution (Table 5). Because the industrial scale adopts each industrial proportion in the total industrial output value, we used the panel data of 20 manufacturing industries from 2000 to 2016 for the regression.

	Energy	Comprehensive pollution	
Variables	$(M_i)$	( y0 <sub>i</sub> )	
$W_{ij}*M_j$	9.400***		
	(2.200)		
$(W_{ij}*M_j)^2$	-9.017***		
	(1.969)		
$M_i$		0.100***	
		(0.029)	
Intra-industry agglomeration	5.593***	0.224	
	(1.160)	(0.177)	
Scale	3.594**	3.290**	
	(1.790)	(1.549)	
Constant	0.795***	-0.097	
	(0.173)	(0.076)	
Controls	Yes	Yes	
R-squared	0.830	0.983	
Year FE	yes	yes	
Industry FE	yes	yes	
Observations	280	280	

Table 5. Impact of intra-industry agglomeration and productive scale on the environment.

Notes: The standard errors are in parentheses; \*\*\*, \*\* and \* represent significance at the 1%, 5% and 10% levels.

The results show that the relationship between inter-industry agglomeration and energy intensity is affected by the intra-industry agglomeration and productive scale; the significance levels are 1% and 5%, respectively. However, the impact of energy intensity on the environment is mainly related to the scale of production, and the relationship with the intra-industry agglomeration is not statistically significant, indicating that the scale of industrial production is the main factor affecting the impact of energy consumption on pollution emissions. Intra-industry agglomeration only affects energy intensity and has no significant direct impact on the pollutant emission level.

Variables	Energy	Comprehensive pollution
variables	$(M_i)$	(y <sub>i</sub> )
$M_i$		0.181***
		(0.022)
$W_{ij}*M_j$	1.882**	
	(0.810)	
$(W_{ij}*M_j)^2$	-2.207***	
	(0.720)	
patents	0.002	-0.009**
	(0.011)	(0.004)
Constant	0.422***	0.138***
	(0.103)	(0.019)
Controls	Yes	Yes
Observations	280	280
R-squared	0.932	0.964
Year FE	yes	yes
Industry FE	yes	yes

Table 6. Effects of technological innovation on environmental pollution.

Notes: The standard errors are in parentheses; \*\*\*, \*\* and \* represent significance at the 1%, 5% and 10% levels.

### 5.3.2. Impact of technological innovation on environmental pollution

We further explore whether agglomeration among industries influences the emission reduction effect of industrial technology innovation. The index of technological innovation uses the number of effective patents, which was taken from the industrial enterprise patent database. The panel data from 2000 to 2016 were used for the regression; the results are shown in Table 6. In the case of agglomeration among different industries (Table 6), there is no significant relationship between industrial innovation and energy intensity; however, industrial innovation was found to have a significant negative impact on environmental pollution, indicating that innovative technology does not reduce the energy consumption of enterprises, but that applying new technology in the process of energy consumption can play a role in reducing emissions. Among these, the regression results in Column (1) show that technological progress is not statistically significant for energy, but that it is positive according to the regression coefficient, indicating that technological progress increases energy intensity to some extent. Technological innovation does not reduce energy demand, as it increases energy demand. Here, we found that the nonlinear effect of inter-industry agglomeration on environmental pollution is clearly weakened, which may be the effects of knowledge and technology spillover.

# 5.4. Heterogeneity analysis

The manufacturing industry can be divided into labor-intensive, capital-intensive and technologyintensive industries. The specific classification is shown in the table below. We further analyzed whether the industry category would influence the effect of inter-industry agglomeration on pollution.

# A. Technology-intensive industry

For technology-intensive industries, Panel A of Table 7 shows that the energy intensity of industry i is related to inter-industry agglomeration and the energy intensity of related industries, and

that the impact of energy intensity varies for different pollutants. The industrial solid waste, industrial sulfur dioxide and carbon emission pollutants of technology-based industries are closely related to energy intensity and increase with increasing energy intensity. Technology-intensive industries, such as the chemical industry, mainly emit SO<sub>2</sub>, and an increase in energy intensity is bound to drive an increase in pollutants. However, the amount of smoke or dust decreased with increasing energy intensity. Due to the inter-industry agglomeration of technology-based industries, such as transportation equipment and electrical machinery industries, the amount of smoke and dust decreased with increased with increase

Variables	Energy ( <i>M</i> <sub>i</sub> )	Comprehensiv e pollution	Wastewater	$SO_2$	Solid waste	Smoke or dust	CO <sub>2</sub>
Panel A: tec	hnology-in	tensive industrie	es				
$W_{ij}*M_j$	-3.437***						
	(1.062)						
$(W_{ij} * M_j)^2$	2.570**						
	(1.238)						
$M_i$		0.050*	-0.016	0.128**	0.532***	-0.037***	0.309***
		(0.027)	(0.024)	(0.052)	(0.034)	(0.012)	(0.017)
Panel B: lab	or-intensiv	e industries					
$W_{ij}*M_j$	-0.970**						
	(0.403)						
$(W_{ij} * M_j)^2$	2.857**						
	(1.178)						
$M_i$		1.091*	7.590***	2.944***	4.299***	-0.421	0.990***
		(0.624)	(1.593)	(0.608)	(0.807)	(0.296)	(0.012)
Panel C: cap	oital-intensi	ve industries					
W <sub>ij</sub> *M <sub>j</sub>	13.21**						
	(5.229)						
$(W_{ij}*M_j)^2$	-18.85**						
	(7.796)						
$M_i$		0.303***	-0.115***	0.765***	2.025***	-0.332*	1.437***
		(0.039)	(0.033)	(0.136)	(0.187)	(0.191)	(0.080)
Controls	yes	yes	yes	yes	yes	yes	yes
Year FE	yes	yes	yes	yes	yes	yes	yes
Industry FE	yes	yes	yes	yes	yes	yes	yes

	Table 7.	Heterogeneity	analysis	of the	inter-industry	<sup>v</sup> agglomeration
--	----------	---------------	----------	--------	----------------	----------------------------

Notes: The standard errors are in parentheses; \*\*\*, \*\* and \* represent significance at the 1%, 5% and 10% levels.

B. Labor-intensive industry

The comprehensive effects of inter-industry agglomeration of labor-intensive industries are shown in Panel B of Table 7. The relationship between inter-industry agglomeration and energy intensity reached a significance level of 5%, and that between the comprehensive level of energy intensity and environmental pollution was found to be significantly positive at 10% level. Therefore, inter-industry agglomeration was found to have a significant impact on environmental pollution via the energy intensity for labor-intensive industries. Combining Columns (1) and (3), the agglomeration of different labor-intensive industries led to an increase in industrial wastewater discharge. The

relationship between wastewater discharge and energy intensity was significantly positive at the 1% level, indicating that labor-intensive industries discharge the greatest amount of industrial wastewater, followed by industrial solid waste, which had a coefficient of 4.299 at a 1% level. In terms of different pollutants, labor-intensive industries have a greater impact on industrial wastewater and solid waste, which involve different pollutants. The production process is accompanied by a large amount of industrial wastewater because labor-intensive industries mainly include the production of textiles, clothing, food, etc. Besides, the biggest characteristic of labor-intensive industries is a large number of people, which would naturally produce household waste, resulting in serious solid waste pollution.

### C. Capital-intensive industry

The energy intensity of capital-intensive industries has a significant impact on the environment (Panel C of Table 7), mainly because industrial solid waste, carbon emissions and sulfur dioxide from capital-intensive industries are more sensitive to energy consumption. The production processes of metal smelting and other manufacturing industries are bound to cause CO<sub>2</sub> emissions to rise, and energy consumption leads to the emission of sulfide and other pollutants, which aggravates environmental pollution. When the industrial solid waste, carbon emissions and sulfur dioxide of capital-intensive industries reach a certain critical value, pollution emissions begin decreasing with increasing energy consumption and the scale effect of inter-industry agglomeration becomes dominant. However, when inter-industry agglomeration exceeds a certain value, due to the expansion of the productive scale, production brings more energy consumption and environmental pollution.

In general, different industries have different effects on environmental pollution. Technologyintensive inter-industry agglomeration has the most significant impact on industrial SO<sub>2</sub>; additionally, the agglomeration of different labor-intensive industries brings industrial wastewater discharge, while capital-intensive industries produce solid waste, carbon emissions and industrial SO<sub>2</sub>. When dealing with different industries, emission reduction measures should be formulated according to the industry type and main pollutants.

### 6. Conclusions

Based on the panel data of 20 Chinese manufacturing industries, we mainly found that there is an inverted U-shaped relationship between inter-industry agglomeration and environmental pollution. At the initial stage, inter-industry agglomeration led to the increase of energy consumption. The deepening of inter-industry agglomeration gradually led to the reduction of energy consumption, which thus reduces environmental pollution. The impact of inter-industry agglomeration on the environment is related to intra-industry agglomeration and production scale. The expansion of the production scale caused by coagglomeration leads to more energy consumption and pollution. Innovation has a significant negative impact on environmental pollution, but not on energy intensity. The innovative technology caused by inter-industry agglomeration does not reduce energy consumption, but it helps to reduce pollution emissions.

The above conclusions provide some insight into global environmental pollution control. As a developing country, China plays an important role in global pollution control. First, we should improve the mechanism of industrial collaborative development. Industrial pollution comes not only from within industries, but also from collaboration between industries with strong industrial linkages. The results in this paper remind us to pay attention to the environmental problem caused by coagglomeration. Second, a reasonable productive scale should be determined according to the degree of coagglomeration to balance economic output and energy consumption. We found that the positive externalities of coagglomeration leads to the "spillover effect" of technology and knowledge, and that

the manufacturing industry would further expand production to obtain more profits. The expansion of the productive scale is an important factor leading to rising energy demand and intensified environmental pollution. Because coagglomeration has inverted U-type characteristics for energy strength, we should determine the productive scale according to the coagglomeration to achieve the purpose of maximum economic output and minimum energy consumption. Finally, energy conservation and emission reduction policies should be linked. The existing policies focus more on emission reduction, and the premise of emission reduction is saving energy. However, energy consumption is the direct factor leading to environmental pollution [102], and China's rapid digital transformation increases industrial energy consumption [103–105]. Simply focusing on emission reduction and neglecting energy conservation can lead to energy consumption increasing, which means that the emission reduction policy would have a negative effect and even aggravate environmental pollution [106].

In this study, we mainly explored whether the agglomeration between industries can help to conserve energy, reduce pollution emission and upgrade technology. However, we did not consider the technical heterogeneity, which can be divided into different types, such as energy efficiency, alternative, transport and waste treatment. In the future, more analysis can be done to verify whether the agglomeration effect brought about by different industries in transportation can save costs or lead to the scale effect. Also, whether different types of technology can have heterogeneous effects on energy intensity and environmental pollution need to be considered in depth in the future.

### **Conflict of interest**

The authors declare that they have no competing interests.

### Appendix A. Supplementary data

Apendix data for this article can be found in the supplementary.

### References

- L. Zhu, Y. Hao, Z. N. Lu, H. Wu, Q. Ran, Do economic activities cause air pollution? Evidence from China's major cities, *Sust. Cities Soc.*, 49 (2019), 1–10. https://doi.org/10.1016/j.scs.2019.101593
- T. C. Li, D. Han, S. Feng, L. Lei, Can Industrial inter-agglomeration between Producer Services and Manufacturing Reduce Carbon Intensity in China?, *Sustainability*, **11** (2019), 1–15. https://doi.org/10.3390/su11154024
- H. Yang, F. Zhang, Y. He, Exploring the effect of producer services and manufacturing industrial co-agglomeration on the ecological environment pollution control in China, *Environ. Dev. Sustain.*, 23 (2021), 16119–16144. https://doi.org/10.1016/j.jclepro.2020.121320
- D. Z. Zeng, L. X. Zhao, Pollution Havens and Industrial Agglomeration, *J. Environ. Econ. Mange.*, 58 (2009), 141–153. https://doi.org/10.1016/j.jeem.2008.09.003
- 5. G. Ellison, E. L. Glaeser, Geographic Concentration in U.S. Manufacturing Industries: A Dartboard Approach, *J. Polit. Econ.*, **105** (1997), 889–927. https://doi.org/10.1086/262098
- Q. X. Gong, G. X. Guo, S. P. Li, X. D. Liang, Examining the Coupling Coordinated Relationship between Urban Industrial inter-agglomeration and Intensive Land Use, *Land*, **10** (2021), 1–12. https://doi.org/10.3390/land10050499

- S. Ke, M. He, C. Yuan, Synergy and Co-agglomeration of Producer Services and Manufacturing: A Panel Data Analysis of Chinese Cities, *Reg. Stud.*, 48 (2014), 1829–1841. https://doi.org/10.1080/00343404.2012.756580
- 8. S. B. Billings, E. B. Johnson, Agglomeration within an Urban area. J. Urban Econ., **91** (2016), 13–25. https://doi.org/10.1016/j.jue.2015.11.002
- Z. Cheng, The Spatial Correlation and Interaction between Manufacturing Agglomeration and Environmental Pollution, *Ecol. Indic.*, 61 (2016), 1024–1032. https://doi.org/10.1016/j.ecolind.2015.10.060
- K. Kanemoto, T. Hanaka, S. Kagawa, K. Nansai, Industrial clusters with substantial carbon–reduction potential, *Econ. Syst. Res.*, **31** (2019), 248–266. https://doi.org/10.1080/09535314.2018.1492369
- 11. L. Brandt, T. Tombe, X. Zhu, Factor market distortions across time, space and sectors in China, *Rev. Econ. Dyn.*, **16** (2013), 39–58. https://doi.org/10.1016/j.red.2012.10.002
- W. Chen, X. Huang, Y. H. Liu, L. Xin, Y. Song, The Impact of High-Tech Industry Agglomeration on Green Economy Efficiency-Evidence from the Yangtze River Economic Belt, *Sustain.*, 11 (2019), 1–18. https://doi.org/10.1108/SAMPJ-09-2018-0248
- M. Song, S. Wang, Can employment structure promote environment-biased technical progress?, *Technol. Forecast. Soc. Change.*, **112** (2016), 285–292. https://doi.org/10.1016/j.techfore.2016.02.016
- L. Ning, F. Wang, J. Li, Urban innovation, regional externalities of foreign direct investment and industrial agglomeration: Evidence from Chinese cities, *Res. Policy.*, 45 (2016), 830–843. https://doi.org/10.1016/j.respol.2016.01.014
- Z. Huang, G. Liao, Z. Li, Loaning scale and government subsidy for promoting green innovation, *Technol. Forecast. Soc. Change.*, 144 (2019), 148–156. https://doi.org/10.1016/j.techfore.2019.04.023
- Z. Li, G. Liao, K. Albitar, Does corporate environmental responsibility engagement affect firm value? The mediating role of corporate innovation, *Bus. Strateg. Environ.*, 29 (2020), 1045–1055. https://doi.org/ 10.1002/bse.2416
- S. J. Li, B. Sun, D. X. Hou, W. J. Jin, Y. Ji, Does Industrial Agglomeration or Foreign Direct Investment Matter for Environment Pollution of Public Health? Evidence From China, *Front. Public Health*, 9 (2021), 1–14. https://doi.org/10.3389/fpubh.2021.711033
- M. W. Wang, R. Gu, M. Wang, J. Zhang, Research on the impact of finance on promoting technological innovation based on the state-space model, *Green Financ.*, 3 (2021), 119–137. https://doi.org/ 10.3934/GF.2021007
- X. Sun, L. Loh, Z. Chen, Effect of market fragmentation on ecological efficiency: evidence from environmental pollution in China, *Environ. Sci. Pollut. Res.*, 27 (2020), 4944–4957. https://doi.org/10.1007/s11356-019-06548-2
- B. Zhang, X. Chen, H. Guo, Doez es central supervision enhance local environmental enforcement? Quasi-experimental evidence from China, J. Public. Econ., 164 (2018), 70–90. https://doi.org/10.1016/j.jpubeco.2018.05.009
- T. Li, X. Li, K. Albitar, Threshold effects of financialization on enterprise R&D innovation: comparison research on heterogeneity, *Quant. Financ. Econ.*, 5 (2021), 496–515. https://doi.org/ 10.3934/QFE.202102

- 22. X.Y. Li, Study on the impact of energy rebound effect on carbon emission reduction at different stages of urbanization in China, *Ecol. Indic.*, **120** (2021), 1–8. https://doi.org/10.1016/j.ecolind.2020.106983
- 23. Z. Li, F. Zou, B. Mo, Does mandatory CSR disclosure affect enterprise total factor productivity? *Ekon. Istraz.*, **35** (2021), 4902–4921. https://doi.org/10.1080/1331677X.2021.2019596
- G. Liao, P. Hou, X. Shen, K. Albitar, The impact of economic policy uncertainty on stock returns: The role of corporate environmental responsibility engagement, *Int. J. Financ. Econ.*, 26 (2021), 4386–4389. https://doi.org/ 10.1002/ijfe.2020
- 25. E. L. Glaser, M. E. Kahn, The Greenness of Cites: Carbon Dioxide Emissions and Urban Development, J. Urban Econ., 67 (2010), 404–418. https://doi.org/10.1016/j.jue. 2009.11.006
- 26. S. Brakman, J. H. Garretsen, R. Gigengack, C. V. Marrewijk, R. Wagenvoort, Negative Feedbacks in the Economy and Industrial Location, J. Reg. Sci., 36 (1996), 631–651. https://doi.org/10.1111/j.1467-9787.1996.tb01122.x
- F. de Leeuw, N. Moussiopoulos, P. Sahm, A. Bartonova, Urban Air Quality in Larger Conurbations in the European Union, *Environ. Modell. Softw.*, 16 (2001), 399–414. https://doi.org/10.1016/S1364-8152(01)00007-X
- W. Ren, Y. Zhong, J. Meligrana, B. Anderson, W. E. Watt, J. Chen, and H. L. Leung, Urbanization, Land Use, and Water Quality in Shanghai:1947–1996. *Environ. Int.*, 29 (2003), 649–659. https://doi.org/10.1016/S0160-4120(03)00051-5
- 29. T. Li, X. Li, G. Liao, Business cycles and energy intensity. Evidence from emerging economies, *Borsa Istanbul Rev.*, **22** (2022), 560–570. https://doi.org/ 10.1016/j.bir.2021.07.005
- B. Wang, X. Nie, Industrial agglomeration and environmental governance: help or resistance--Evidence from quasi natural experiment in development zones, *China Ind. Econ.*, **12** (2016), 75– 89. (In Chinese). https://doi.org/ 10.19581/j.cnki.ciejournal.2016.12.006
- S. X. Liu, Y. M. Zhu, K. Q. Du, The impact of industrial agglomeration on industrial pollutant emission: evidence from China under New Normal, *Clean Technol. Environ. policy*, **19** (2017), 2327–2334. https://doi.org/10.1007/s10098-017-1407-0
- 32. V. Erik, N. Peter, Urban Environmental Externalities, Agglomeration Forces, and the Technological 'Deus ex Machina', *Environ. Plan. A.*, **40** (2008), 928–947. https://doi.org/10.1068/a38434
- C. C. Fan, A. J. Scott, Industrial Agglomeration and Development: A Survey of Spatial Economic Issues in East Asia and a Statistical Analysis of Chinese Regions, *Econ. Geogr.*, **79** (2003), 295– 319. https://doi.org/10.1111/j.1944-8287.2003.tb00213.x
- J. Lan, M. Kakinaka, X. Huang, Foreign Direct Investment, Human Capital and Environmental Pollution in China, *Environ. Resour. Econ.*, **51** (2012), 255–275. https://doi.org/10.1007/s10640-011-9498-2
- 35. Y. Zhu, Y. Xia, The Impact of Industrial Agglomeration on Environmental Pollution: Evidence from China under New Urbanization, *Energ. Environ.*, **30** (2018), 1010–1026. https://doi.org/10.1177/0958305X18802784
- Y. L. Ye, S. Ye, H. C. Yu, Can Industrial Collaborative Agglomeration Reduce Haze Pollution? City-Level Empirical Evidence from China, *Int. J. Environ. Res. Public Health*, 18 (2021), 1566– 1585. https://doi.org/10.3390/ijerph18041566
- J. Gorelick, N. Walmsley, The greening of municipal infrastructure investments: technical assistance, instruments, and city champions, *Green Financ.*, 2 (2020), 114–134. https://doi.org/ 10.3934/GF.2020007

- 38. K. Tanaka, S. Managi, Industrial agglomeration effect for energy efficiency in Japanese production plants, *Energ. Policy*, **156** (2021). https://doi.org/10.1016/j.enpol.2021.112442
- S. Naik, V. Bagodi, Energy conservation opportunities: evidences from three industrial clusters in India, *Int. J. Energy Sect. Manag.*, 15 (2021), 600–627. https://doi.org/10.1108/IJESM-07-2020-0022
- M. Yi, Y.Q. Wang, M. Y. Sheng, B. Sharp, Y. Zhang, Effects of heterogeneous technological progress on haze pollution: Evidence from China, *Ecol. Econ.*, 169 (2020), 106533–106544. https://doi.org/10.1016/j.ecolecon.2019.106533
- 41. R. Tveteras, G. E. Battese, Agglomeration Externalities, Productivity and Technical Inefficiency, *J. Reg. Sci.*, **46** (2006), 605–625. https://doi.org/10.1111/j.1467-9787.2006.00470.x
- 42. A. Ciccone, R. E. Hall, Productivity and the Density of Economic Activity, *Am. Econ. Rev.*, **86** (1996), 1–39. http://www.jstor.org/stable/2118255
- J. H. Zhu, Z. H. Li, Can digital financial inclusion effectively stimulate technological Innovation of agricultural enterprises? --A case study on China, *Natl. Account. Rev.*, 3 (2021), 398–421. https://doi.org/ 10.3934/NAR.2021021
- M. X. Wang, L. Li, H. Y. Lan, The measurement and analysis of technological innovation diffusion in China's manufacturing industry, *Natl. Account. Rev.*, 3 (2021), 452–471. https://doi.org/ 10.3934/NAR.2021024
- 45. Z. Li, H. Chen, B. Mo, Can digital finance promote urban innovation? Evidence from China, *Borsa. Istanbul Rev.*, 2022, 1–12. https://doi.org/10.1016/j.bir.2022. 10.006
- 46. A. Martin, P. L. Johan, W. Joakim, The economic micro geography of diversity and specialization externalities-firm-level evidence from Swedish cities, *Res. Policy*, **48** (2019), 1385–1398. https://doi.org/10.1016/j.respol.2019.02.003
- 47. Q. Y. Zheng, B. Q. Lin, Impact of industrial agglomeration on energy efficiency in China's paper industry, *J. Clean. Prod.*, **184** (2019), 1072–1080. https://doi.org/10.1177/0958305X18802784
- M. Tahir, E. Ahmad, The relationship of energy intensity with economic growth: Evidence for European economies, *Energy Strategy Rev.*, 20 (2018), 90–98. https://doi.org/10.1016/j.esr.2018.02.002
- M. He, B. Walheer, Technology intensity and ownership in the Chinese manufacturing industry: A labor productivity decomposition approach, *Natl. Account. Rev.*, 2 (2020), 110–137. https://doi.org/ 10.3934/NAR.2020007
- 50. Y. Liu, P. Failler, Y. Ding, Enterprise financialization and technological innovation: Mechanism and heterogeneity, *PLoS One*, **17** (2022), 1–21. https://doi.org/10.1371/journal.pone.0275461
- 51. X. H. Liu, Dynamic evolution, spatial spillover effect of technological innovation and haze pollution in China, *Energy Environ.*, **29** (2018), 968–988.
- 52. L. M. Ma, X. Zhang, Spatial effects of China's haze and its impact on economic and energy structure, *China Ind. Econ.*, **4** (2014), 19–31.
- C. Hua, J. J. Miao, W. P. Liu, G. Du, X. Wang, The impact mechanism of industrial agglomeration on energy efficiency-Evidence from producer service industry in China, *Energ. Sources Part B*, 16 (2021), 740–758
- M. M. Rahman, K. Alam, The nexus between health status and health expenditure, energy consumption and environmental pollution: empirical evidence from SAARC-BIMSTEC regions, *BMC Public Health*, **21** (2021), 1–12. https://doi.org/10.21203/rs.3.rs-68393/v2

- 55. X. Sun, B. K. Zhu, S. Zhang, H. Zeng, K. Li, B. Wang, Z. F. Dong, C. C. Zhou, New indices system for quantifying the nexus between economic-social development, natural resources consumption, and environmental pollution in China during 1978–2018, *Sci. Total Environ.*, 804 (2021), 150180. https://doi.org/10.1016/j.scitotenv.2021.150180
- D. Federica, C. Antonio, B. M. Silvio Servitization and sustainability actions, Evidence from European manufacturing companies, *J. Environ. Manage.*, 234 (2019), 367–378. https://doi.org/10.1016/j.jenvman.2019.01.004
- 57. M. M. Faisal, K. Afra, Energy consumption, carbon emissions and economic growth in Pakistan: Dynamic causality analysis, *Renew. Sust. Energ. Rev.*, **72** (2016), 1233–1240. https://doi.org/10.1016/j.rser.2016.10.081
- C. Huang, J. W. Wang, C. M. Wang, J. H. Cheng, J. Dai, Does tourism industry agglomeration reduce carbon emissions? *Environ. Sci. Pollut. Res.*, 28 (2021), 30278–30293. https://doi.org/10.1007/s11356-021-12706-2
- Z. Huang, H. Dong, S. Jia, Equilibrium pricing for carbon emission in response to the target of carbon emission peaking, *Energy Econ.*, **112** (2022), 106160–106175. https://doi.org/10.1016/j.eneco.2022.106160
- 60. X. Y. Li, H. Z. Xu, The Energy-conservation and Emission-reduction Paths of Industrial sectors: Evidence from China's 35 industrial sectors, *Energy Econ.*, **86** (2020), 10462–10476. https://doi.org/10.1016/j.eneco.2019.104628
- 61. P. G. Saculsan, T. Kanamura, Examining risk and return profiles of renewable energy investment in developing countries: the case of the Philippines, *Green Financ.*, **2** (2019), 135–150. https://doi.org/ 10.3934/GF.2020008
- 62. O. Sukharev, V. Ekaterina, Financial and non-financial investments: comparative econometric analysis of the impact on economic dynamics, *Quant. Financ. Econ.*, **4** (2020), 382–411. https://doi.org/ 10. 3934/QFE. 2020018
- 63. G. Carvalho, P. Roberto, S. Joelson, Venture capital backing: financial policies and persistence over time, *Quant. Financ. Econ.*, **5** (2021), 640–663. https://doi.org/ 10.3934/QFE.2021029
- T. Li, J. Zhong, Z. Huang, Potential Dependence of Financial Cycles between Emerging and Developed Countries: Based on ARIMA-GARCH Copula Model, *Emerg. Mark. Finance Trade*, 56 (2019), 1237–1250. https://doi.org/10.1080/1540496x.2019.1611559
- 65. Y. Liu, Z. Li, M. Xu, The influential factors of financial cycle spillover: evidence from China, *Emerg. Mark. Financ. Trade*, 56 (2020), 1336–1350. https://doi.org/10.1080/1540496x.2019.1658076
- 66. Z. Li, C. Yang, Z. Huang, How does the fintech sector react to signals from central bank digital currencies?, *Finance Res. Lett.*, 50 (2022), 103308–103313. https://doi.org/10.1016/j.frl.2022.103308
- Y. Wang, S. Yin, X. Fang, W. Chen, Interaction of economic agglomeration, energy conservation and emission reduction: Evidence from three major urban agglomerations in China, *Energy*, 241 (2022), 122519–122532. https://doi.org/10.1016/j.energy.2021.122519
- H. Baydoun, M, Age, The Effect of Energy Consumption and Economic Growth on Environmental Sustainability in the GCC Countries: Does Financial Development Matter?, *Energies*, 14 (2021), 5897–5915. https://doi.org/10.3390/en14185897

- H. X. Liu, K. R. Du, J. Li, An improved approach to estimate direct rebound effect by incorporating energy efficiency: A revisit of China's industrial energy demand, *Energ. Econ.*, 80 (2019), 720–730. https://doi.org/10.1016/j.eneco.2019.02.012
- Y. J. Zhang, H. R. Peng, B. Su, Energy rebound effect in China's Industry: An aggregate and disaggregate analysis, *Energ. Econ.*, 61 (2017), 199–208. https://doi.org/10.1016/j.eneco.2016.11.011
- M. M. Xu, B. Q. Lin, S. Q. Wang, Towards energy conservation by improving energy efficiency? Evidence from China's metallurgical industry. *Energy*, **216** (2021), 119255–119355. https://doi.org/10.1016/j.energy.2020.119255
- 72. Y. Ushifusa, A. Tomohara, Productivity and Labor Density: Agglomeration Effects over Time, *Atl. Econ. J.*, **41** (2013), 123–132. https://doi.org/10.1007/s11293-012-9354-y
- 73. P. H. Berkhout, J. C. Muskens, J. W. Velthuijsen, Defining the Rebound Effect, *Energy Policy*, **28** (2000), 425–432. https://doi.org/10.1016/S0301-4215(00)00022-7
- S. R. Mudakkar, K. Zamank, M. M. Khan, M. Ahmad, Energy for Economic Growth, Industrialization, Environment and Nature Resources: Living Is Just Enough, *Renew. Sust. Energ. Rev.*, 25 (2013), 580–595. https://doi.org/10.1016/j.rser.2013.05.024
- D. Liu, B. Xiao, Can China achieve its carbon emission peaking? A scenario analysis based on STIRPAT and system dynamics model, *Ecol. Indic.*, **93** (2018), 647–657. https://doi.org/10.1016/j.ecolind.2018.05.049
- 76. Y. L. Chen, Z. Wang, Z. Q. Zhong, CO2 emissions, economic growth, renewable and non-renewable energy production and foreign trade in China, *Renew. Energ.*, **131** (2019), 208–216. https://doi.org/10.1016/j.renene.2018.07.047
- 77. Q. Zhang, R. Brouwer, Is China Affected by the Resource Curse? A Critical Review of the Chinese Literature, *J. Policy Model*, **42** (2020), 133–152. https://doi.org/ 10.1016/j.jpolmod.2019.06.005
- S. Chen, Y. Wang, K. Albitar, Z. Huang, Does ownership concentration affect corporate environmental responsibility engagement? The mediating role of corporate leverage, *Borsa Istanbul Rev.*, 21 (2021), 13–24. https://doi.org/ 10.1016/j.bir.2021.02.001
- J. Hou, Y. Hou, Q. Wang, N. Yue, Can industrial agglomeration improve energy efficiency? Empirical evidence based on China's energy-intensive industries, *Environ. Sci. Pollut. Res.*, 29 (2022), 80297–80311. https://doi.org/10.1007/s11356-022-21429-x
- C. Fan, A. J. Scott, Industrial Agglomeration and Development: A Survey of Spatial Economic Issues in East Asia and A Statistical Analysis of Chinese Regions, *Econ. Geogr.*, **79** (2003), 295– 319. https://www.tandfonline.com/doi/abs/10.1111/j.19448287.2003.tb00213.x
- 81. M. E. Porter, C. Linde, Toward a New Conception of the Environment- Competitiveness Relationship, J. Econ. Perspect, 9 (1995), 97–118. https://doi.org/10.1257/jep.9.4.97
- 82. Y. Su, Z. Li, C. Yang, Spatial interaction spillover effects between digital financial technology and urban ecological efficiency in China: an empirical study based on spatial simultaneous equations, *Int. J. Environ. Res. Public Health*, **18** (2021), 8535–8561. https://doi.org/10.3390/ijerph18168535
- 83. H. Zhao, B. Lin, Will agglomeration improve the energy efficiency in China's textile industry: Evidence and policy implications. *Appl. Energy*, 237 (2018), 326–337. https://doi.org/10.1016/j.apenergy.2018.12.068

- J. Wang, J. Ma, Has tourism industry agglomeration improved the total factor productivity of Chinese urban agglomerations?--The moderating effect of public epidemic, *Front. Public Health*, 525 (2022), 854681–854691. https://doi.org/10.3389/fpubh.2022.854681
- 85. J. He, Pollution haven hypothesis and environmental impacts of foreign direct investment: The case of industrial emission of sulfur dioxide (SO2) in Chinese provinces, *Ecol. Econ.*, **60** (2006), 228–245. https://doi.org/10.1016/j.ecolecon.2005.12.008
- H. Wu, Y. Hao, S. Ren, How do environmental regulation and environmental decentralization affect green total factor energy efficiency: Evidence from China, *Energy Econ.*, **91** (2020), 104880–104898. https://doi.org/10.1016/j.eneco.2020.104880
- 87. K. Dong, M. Shahbaz, J. Zhao, How do pollution fees affect environmental quality in China? *Energ. Policy*, **160** (2022), 112695–112709. https://doi.org/10.1016/j.enpol.2021.112695
- Z. Yang, Z. Xiong, W. Xue, Y. Zhou, The Impact of Pollution Fee Reform on the Emission of Water Pollutants: Evidence from Manufacturing Enterprises in China, *Int. J. Environ. Res. Public Health*, 19 (2022), 10660–10677. https://doi.org/10.3390/ijerph191710660
- 89. C. Qu, J. Shao, Z. Shi, Does financial agglomeration promote the increase of energy efficiency in China? *Energy Policy*, **146** (2020), 111810–111825. https://doi.org/10.1016/j.enpol.2020.111810
- F. Han, R. Xie, J. Y. Fang, Y. Liu, The effects of urban agglomeration economies on carbon emissions: Evidence from Chinese cities, *J. Clean. Prod.*, **172** (2018), 1096–1110. https://doi.org/10.1016/j.jclepro.2017.09.273
- H. Xie, Q. Chen, F. Lu, W. Wang, G. Yao, J. Yu, Spatial-temporal disparities and influencing factors of total-factor green use efficiency of industrial land in China, *J. Clean. Prod.*, 207 (2019), 1047–1058. https://doi.org/10.1016/j.jclepro.2018.10.087
- X. Li, X. Lai, F. Zhang, Research on green innovation effect of industrial agglomeration from perspective of environmental regulation: Evidence in China, J. Clean. Prod., 288 (2021), 125583– 125595. https://doi.org/10.1016/j.jclepro.2020.125583
- 93. S. Sorrell, J. Dimitropoulos, The rebound effect: Microeconomic definitions, limitations and extensions, *Ecol. Econ.*, **65** (2008), 636–649. https://doi.org/10.1016/j.ecolecon.2007.08.013
- S. Shao, K. Zhang, J. M. Dou, Energy conservation and emission reduction effect of economic agglomeration: Theory and Chinese experience, *Manage. World*, **35** (2019), 36–60. (In Chinese). https://doi.org/ 10.19744/j.cnki.11-1235/f.2019.0005.
- 95. T. Jaynes, On the rationale of maximum-entropy methods, *Proc. IEEE. Inst. Electr. Electron Eng.*, 70 (1982), 939–952. https://doi.org/ 10.1109/PROC.1982.12425
- B. Q. Lin, and R. P. Tan, Chinese Economic Cluster and Green Economic Efficiency, *Econ. Res.*, 54 (2019), 119–132. (In Chinese)
- 97. H. Abdi, L. J. Williams, Principal component analysis, *Wiley interdiscip. Rev. comput. Stat.*, **2** (2010), 433–459. https://doi.org/10.1002/wics.101
- Y. M. Zhu, S. X. Liu, Y. J. Li, Y. Pei, H. Q. Qiao, The mitigation effect of industrial agglomeration on environmental pollution: the theory and the empirical evidence, *Environ. Econ. Res.*, 4 (2019), 86–107. (In Chinese). https://doi.org/10.19511/j.cnki.jee.2019.01.007
- T. H. Chang, P. J. Klenow, Misallocation and Manufacturing TFP in China and India, Q. J. Econ., 124 (2009), 1403–1448. https://doi.org/10.1162/qjec.2009.124.4.1403
- 100.H. W. Wen, C. C. Lee, Z. Y. Song, Digitization and environment: how does ICT affect enterprise environmental performance?, *Environ. Sci. Pollut. Res.*, **28** (2021), 54826–54841. https://doi.org/10.1007/s11356-021-14474-5

- 101.X. E. Qu, Empirical Analysis of Comprehensive Evaluation of Environmental Pollution in China, *Ind. Econ. Res.*, **04** (2014), 51–102. (In Chinese). https://doi.org/10.13269/j.cnki.ier.2014.04.006
- 102.Y. Sun, Y. Li, T. Yu, X. Zhang, L. Liu, P. Zhuang, Resource extraction, environmental pollution and economic development: Evidence from prefecture-level cities in China, *Resour. Policy*, 74 (2021), 102330–102342. https://doi.org/10.1016/j.resourpol.2021.102330
- 103.Z. H. Li, J. H. Zhu, J. J. He, The effects of digital financial inclusion on innovation and entrepreneurship: A network perspective, *Electron. Res. Arch.*, **30** (2022), 4697–4715. https://doi.org/10.3934/era.2022238
- 104.T. H. Li, J. Y. Wen, D. W. Zeng, K. Liu, Has enterprise digital transformation improved the efficiency of enterprise technological innovation? A case study on Chinese listed companies, *Math. Biosci. Eng.*, **19** (2022), 12632–12654. https://doi.org/10.3934/mbe.2022590
- 105.Y. Liu, P. Failler, Y. Ding, Enterprise financialization and technological innovation: Mechanism and heterogeneity, *PLoS One*, **17** (2022), 1–21. https://doi.org/10.1371/journal.pone.0275461
- 106.Y. Liu, P. Failler, Z. Y. Liu, Impact of Environmental Regulations on Energy Efficiency: A Case Study of China's Air Pollution Prevention and Control Action Plan, *Sustainability*, 14 (2022), 1– 21. https://doi.org/10.3390/su14063168



©2023 the Author(s), licensee AIMS Press. This is an open access article distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/4.0)